Author's response to reviews

Title: Tamoxifen Elicited Uterotrophy: Cross-Species and Cross-Ligand Analysis of the Gene Expression Program

Authors:

Joshua C Kwekel (joshua.kwekel@gmail.com)
Agnes L Forgacs (aforgacs@msu.edu)
Lyle D Burgoon (burgoonl@msu.edu)
Kurt J Williams (will1273@msu.edu)
Tim R Zacharewski (tzachare@msu.edu)

Version: 5 Date: 27 March 2009

Author's response to reviews:

-references to all additional files have been placed in the text
-duplicate reference section has been deleted
-all figures have been checked for cropping, please advise if there is a particular figure that requires more attention.

thank you.